Author's response to reviews

Title: A phase II experience with neoadjuvant irinotecan (CPT-11), 5-fluorouracil (5-FU) and leucovorin (LV) for colorectal liver metastases

Authors:

Oliver F Bathe (bathe@ucalgary.ca)
Scott Ernst (escott45@uwo.ca)
Francis R Sutherland (francis.sutherland@calgaryhealthregion.ca)
Elijah Dixon (elijah.dixon@calgaryhealthregion.ca)
Charles Butts (charlesb@cancerboard.ab.ca)
David Bigam (dbigam@cha.ab.ca)
David Holland (dr-holland@shaw.ca)
Geoffrey A Porter (geoff.porter@dal.ca)
Jennifer Koppel (jskoppel@ucalgary.ca)
Scot Dowden (scotdowd@cancerboard.ab.ca)

Version: 4 Date: 17 May 2009

Author's response to reviews:

Dear Dr. Marshall

The following revisions have been made as per the emailed instructions from May 6, 2009:

1. Affiliations have been written in the title page in the prescribed format.
2. All vertical elements have been removed from the tables. The table title has been formatted properly.
3. The figure title has been removed from all figures, and figures have been cropped.
4. The title of the manuscript has been converted to sentence case in the manuscript file and in the submission system.
5. The running head has been removed from the title page.
6. The research support that had appeared on the title page was moved to the Acknowledgements section.
7. Email addresses were included on the title page in the prescribed format.
8. All underlining has been removed.
9. The Introduction section in the abstract and the manuscript has been renamed as "Background".
10. Materials and Methods has been renamed as "Methods"
11. The manuscript has been reviewed for any typos.

Thank you for the opportunity of publishing in BMC Cancer.
Sincerely,
Oliver Bathe